Author's response to reviews

Title: Difference of the associations between self-rated health and demographic characteristics, lifestyle, and psychosocial work environment between two representative types of Chinese worksite

Authors:

Yingnan Jia (12111020010@fudan.edu.cn)
Junling Gao (jlgao@fudan.edu.cn)
Junming Dai (jmdai@shmu.edu.cn)
Pinpin Zheng (zpinpin@shmu.edu.cn)
Xiaoyu Wu (wuxiaoyu@shaw.gov.cn)
Guangyao Li (liguangyao@shaw.gov.cn)
Hua Fu (hfufudan.edu.cn)

Version: 2
Date: 14 March 2014

Author's response to reviews:

The manuscript has been revised according to the requests. We apologize for the delay. We hope this paper is suitable for “BMC Public Health”